

Efficient Handling of String-Number Conversion

Parosh Aziz Abdulla Uppsala University Uppsala, Sweden parosh@it.uu.se

Bui Phi Diep Uppsala University Uppsala, Sweden bui.phi-diep@it.uu.se

Hsin-Hung Lin Academia Sinica Taipei, Taiwan hlin@iis.sinica.edu.tw Mohamed Faouzi Atig Uppsala University Uppsala, Sweden mohamed_faouzi.atig@it.uu.se

> Julian Dolby IBM Research NY, USA dolby@us.ibm.com

Lukáš Holík Brno University of Technology Brno, Czechia holik@fit.vutbr.cz Yu-Fang Chen Academia Sinica Taipei, Taiwan yfc@iis.sinica.edu.tw

Petr Janků Brno University of Technology Brno, Czechia ijanku@fit.vutbr.cz

Wei-Cheng Wu University of Southern California CA, USA wwu@isi.edu

Abstract

String-number conversion is an important class of constraints needed for the symbolic execution of stringmanipulating programs. In particular solving string constraints with string-number conversion is necessary for the analysis of scripting languages such as JavaScript and Python, where string-number conversion is a part of the definition of the core semantics of these languages. However, solving this type of constraint is very challenging for the state-of-the-art solvers. We propose in this paper an approach that can efficiently support both string-number conversion and other common types of string constraints. Experimental results show that it significantly outperforms other state-of-the-art tools on benchmarks that involves string-number conversion.

CCS Concepts: • Security and privacy \rightarrow Logic and verification; • Software and its engineering \rightarrow Formal methods.

Keywords: String Solver, Formal Verification, Automata

ACM Reference Format:

Parosh Aziz Abdulla, Mohamed Faouzi Atig, Yu-Fang Chen, Bui Phi Diep, Julian Dolby, Petr Janků, Hsin-Hung Lin, Lukáš Holík, and Wei-Cheng Wu. 2020. Efficient Handling of String-Number Conversion. In *Proceedings of the 41st ACM SIGPLAN International*

PLDI '20, June 15-20, 2020, London, UK

© 2020 Association for Computing Machinery. ACM ISBN 978-1-4503-7613-6/20/06...\$15.00 https://doi.org/10.1145/3385412.3386034 Conference on Programming Language Design and Implementation (PLDI '20), June 15–20, 2020, London, UK. ACM, New York, NY, USA, 15 pages. https://doi.org/10.1145/3385412.3386034

1 Introduction

Symbolic execution is a very popular technique that allows programmers to check the feasibility of a path in a program, i.e., determining the value of the inputs under which the given path can be executed. The path feasibility problem is usually solved by a reduction to the satisfiability of a formula. More precisely, program statements in the path are translated to equivalent constraints in static single assignment (SSA) form and then solved by *Satisfiability Modulo Theory (SMT)* solvers. The types of constraints needed depend on the types of program expressions to be analyzed. Therefore, SMT solvers need to support different combinations of theories so that they can handle a wide range of types.

Among all data types, the *string data type* is omnipresent in modern programming languages. Various security vulnerabilities such as injection and cross-site scripting attack are caused by malicious string values. Therefore, string constraint solving has received considerable attention in the constraint solving community. Operations such as *equality constraints* (e.g. x.y = y.x), *regular constraints* (e.g., $x \in (a.b)^*$), and *integer constraints* (e.g., |x| - |y| > 3), are widely supported by most state-of-the-art string constraint solvers such as, CVC4 [8], OSTRICH [13], Sloth [21], Trau+ [1, 2, 5], Z3 [15] and Z3Str3 [9].

An important class of string operations is the stringnumber conversions. While string length operations are sometimes well supported, converting a string x to an integer n (e.g., using the operation n = toNum(x)) or turning an integer value n into its string form x (e.g., using the operation x = toStr(n)) suffer from limited support (in terms of the scale of formulae they can handle) by the state-of-the-art string constraint solvers.

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than ACM must be honored. Abstracting with credit is permitted. To copy otherwise, or republish, to post on servers or to redistribute to lists, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

}

In fact, a code that receives string input tends to need to convert at least some of that input into numbers. For example, the program fragment below is a variant of the Luhn test algorithm that is often used in credit card or ID validation.

```
function checkLuhn(value) {
var sum = 0;
for (var i = value.length - 1; i >= 0; i-=2) {
    var d = parseInt(value.charAt(i));
    sum += d;
}
for (var i = value.length - 2; i >= 0; i-=2) {
    var d = parseInt(value.charAt(i));
    if ((d *= 2) > 9) d == 9;
    sum += d;
}
var last= sum.toString().charAt(sum.length-1);
return last == '0';
```

The input value of the Luhn test algorithm is a sequence of digits. The algorithm processes the digits in the reversed order. The value of every odd digit (e.g., 1st, 3rd, etc.) is added to sum directly. For the value of every even digit, the algorithm (1) doubles its value, (2) subtracts its value by 9 if the doubled-value is larger than 9, and (3) adds the final result to sum. At the end, the input is validated if the last digit of sum is 0 (i.e., sum mod 10=0).

To check whether the program path that traverses both loops exactly once and finally passes this test has a valid input, we create the following (string) constraint:

1	$value_0 \in [1, 9]^+ \land sum_0 = 0 \land$
2	$i_0 = value_0 - 1 \wedge$
3	$d_0 = \text{toNum}(\text{charAt}(value_0, i_0)) \land$
4	$sum_1 = sum_0 + d_0 \wedge$
5	$i_1 = value_0 - 2 \wedge$
6	$d_1 = \text{toNum}(\text{charAt}(value_0, i_1)) \land$
7	$sum_2 = sum_1 + ite(d_1 * 2 > 9, d_1 * 2 - 9, d_1 * 2) \land$
8	$i_2 = 0$
9	$last_0 = charAt(toStr(sum_2), toStr(sum_2) - 1) \land$
10	$last_0 = "0"$

Here $value_0$ and $last_0$ are string variables and the others are integer variables. The method charAt(x, i) returns the character at index *i* in the string *x* while n = ite(b, e, e')assigns to *n* the value of the expression *e* if *b* is true and the value of the expression e' otherwise. Line 1 describes the initial condition: value should be a sequence of digits and sum is initially zero. Lines 2-4 and lines 5-7 describe one execution of the first and second loop, respectively. Line 8 describes the condition on i_2 before leaving the loop. Finally, Lines 9-10 describe the condition that the last digit of *sum* is zero. Observe that to describe such a program path, we need a solver that supports the following types of constraints:

• *Regular* constraints (e.g., $value_0 \in [1, 9]^+$, which says *value*₀ is in the regular language $[0, 9]^+$),

- Integer constraints (e.g., $i_0 = |value_0| 1$, which says i_0 equals the length of *value*₀ minus one),
- Equality constraints (often y = charAt(x, i) is encoded as $x = x_1 \cdot x_2 \cdot x_3 \wedge |x_1| = i \wedge |x_2| = 1 \wedge y = x_2$, which uses equality of string terms x and $x_1.x_2.x_3$), and
- String-number conversion (e.g., toStr(sum₂), which is the string value of the number sum_2).

Most of the state-of-the-art string constraint solvers provide limited support to the combination of above constraints. In Table 3 of our evaluation (Section 9), CVC4 fails to solve constraints corresponding to checkLuhn of more than 6 loop iterations in 2 minutes, Z3 can only solve the cases corresponding to 2 to 5, and 9 loop iterations, and Z3Str3 fails to solve any case.

Even, more crucially, in many programming languages, string-number conversion is a part of the definition of their core semantics. JavaScript, which powers most interactive content on the Web and increasingly server-side code with Node.js, is one of such languages. Other scripting languages do too, but we focus on JavaScript due to its prominence. To see how string-integer conversion pervades semantics, consider the following program:

A casual glance at the above code reveals no use of strings at all, but the semantics of field access is somewhat unusual in JavaScript. In fact, the arrays in JavaScript are indexed by strings, and numeric indices are converted to strings. This conversion is mandated explicitly by the JavaScript semantics. The 2019 edition of ECMAScript [16] requires that ToPropertyKey be called on the element expression (§12.3.2.1), and *ToPropertyKey* calls ToString on that value in all but special cases (§7.1.14). Therefore, any faithful symbolic execution of JavaScript must handle such conversions for even basic array operations to work correctly. Consider the following code snippet that manipulates an array x, with its value shown on the right:

1	x = [0, 0, 0, 0, 0]	[0,0,0,0,0]
2	x[3] = 4	[0,0,0,4,0]
3	x[03] = 2	[0,0,0,2,0]
4	x["3"] = 5	[0,0,0,5,0]
5	x["03"] = 7	[0,0,0,5,0] and x["03"] = 7
6	x["03"-1] = 2	[0,0,2,5,0] and x["03"] = 7

Here x[3] in line 2, x[03] in line 3, and x["3"] in line 4 all denote the same array element of x["3"] (due to the implicit conversion of numeric indices to strings in JavaScript), but x["03"] denotes a completely different element (which is stored at the index "03" of the array). So naïve modeling of array indices with integers will not work - it cannot distinguish the indices "3" and "03".

But if array indices are modeled as strings, we must handle arithmetic somehow. Let us look at the case of line 6, we need to update the value of x["03"-1]. The evaluation of the expression "03"-1 involves an implicit type conversion from the string "03" to an integer value 3 due to the - (minus) operation. The result of the evaluation of "03"-1 is the integer 2, which is then converted back to string "2" and used as the array index. Hence x["03"-1] means the array element of x["2"]. Even for a simple example like this, the conversion between string and number is unavoidable. This is a rather basic array operation in JavaScript, and not handling string-number conversion operations will cripple any analysis of non-trivial JavaScript code. Thus, we need stronger solvers that are able to handle string-number conversion operations in order to be able to analyze real code.

Solving string constraint with string-number conversion is a very challenging problem. From the theoretical point of view, this problem is already proven to be undecidable [14]. From practical point of view, our experimental results (in Section 9) show that the current the state-of-the-art string constraint solvers provide little support to string-number conversion.

In this paper, we propose a framework that efficiently handles string constraints with string-number conversion. Since the problem is provably unsolvable, our framework combines over and under-approximation techniques. The over-approximation is for proving UNSAT when possible, while the under-approximation is for proving SAT when possible. Both over- and under-approximation fall in a decidable fragment of string constraints that we can efficiently solve.

For ease of presentation, we use the following toy example

 $\Phi = \{ 0^{\circ}x = x^{\circ}0^{\circ}, toNum(x) = toNum(y), |y| > |x| > 1, 1000 < |y| \}$

to explain the main ideas behind our decision procedure. To make our terminology explicit: Φ states that "0" concatenated with x is the same as x with "0", the numeric value of the string x is equivalent to that of y, y is longer than x, y is longer than 1000 characters, and x is longer than 1. Notice that Φ is satisfiable. E.g., it has a model x = "00" and y = "0¹⁰⁰²". Although this toy example is seemingly trivial, all the state-of-the-art string constraint solvers we tried (including Z3, CVC4, and Z3Str3) cannot solve it within 10 minutes.

Our new decision procedure solves the example in few seconds. It proceeds in two steps: The first step consists in over-approximating the set of input constraints into a set that falls in the chain-free fragment [5], which is decidable. Observe that we could over-approximate the input constraint into any decidable fragment, e.g. the acyclic fragment [3] or the straight-line fragment [13]. Our choice of the chain-free fragment [5] is only motivated by the fact that the chain-free fragment is the *largest* known decidable fragment for that class of string constraints. In our example, we overapproximate the formula Φ by converting "0"x = x "0" to two formulae { $x_1 = "0$ " $x, x_2 = x$ "0"} and replacing the constraint toNum(x) = toNum(y) with $n_x = n_y \land (n_x = -1 \lor (n_x \neq -1 \land x \in [0-9]^*)) \land (n_y = -1 \lor (n_y \neq -1 \land y \in [0-9]^*)))$. Observe



Figure 1. An example of a parametric flat automaton

that if the over-approximation is UNSAT then our decision procedure declares that the original formula is also UNSAT and terminates. Surprisingly, despite its simplicity, our overapproximation procedure works very well in practice as shown by our experimental results (in Section 9). Coming back to the formula Φ , the over-approximation module will return SAT in this case.

The second step of our decision procedure is only enabled if the over-approximation step returns SAT. In this case, our decision procedure uses an under-approximation technique (which is our main contribution) to restrict the search domain of each string variable to strings that obey some predefined and parameterized pattern. We propose to use patterns defined by parametric flat automata (PFA). A PFA is a *flat* finite state automaton consisting of a predefined sequence of loops, each of fixed length (see Figure 1). The size of the PFA is parameterized by the length of the sequence of loops and the size of each loop. Adjusting these parameters enlarges or prunes the potential solution space. This approach based on PFA is very flexible yet allows very efficient manipulation. In fact, our procedure restricts the search space for each variable to the set of words accepted by the corresponding given PFA.

Then, we show that given such restriction, one can reduce the string constraint solving problem to a linear formula satisfiability problem in polynomial-time. To gain in efficiency, we label each transition of a PFA with a unique *character* variable (whose domain is the set of natural numbers) instead of having a transition between every two states for each symbol in the alphabet. This is done by associating to each character in our alphabet a unique natural number. This allows us to avoid the *alphabet explosion problem* from which the approach in [1] suffers and it is the key for handling string-number conversion efficiently.

In the following, we explain the construction of the linear formula using Φ as an example. Assume that we project the domains of *x* and *y* to the PFA in Figure 2 (a) and (b), respectively. The variables v_0 , v_1 , v_2 , v_3 in the figure are *character* variables. Thus, v_0 , v_1 , v_2 , v_3 are also integer variables.

The linear formula produced after the domain restriction will be over variables v_0 , v_1 , v_2 , v_3 , as well as the number of occurrences of each character variable $\#v_0$, $\#v_1$, $\#v_2$, $\#v_3$. Each model of the linear formula encodes a model of the string constraint. For example, x = "00" and $y = "0^{1002"}$ is encoded by the assignment (v_0 , v_1 , v_2 , v_3 , $\#v_0$, $\#v_1$, $\#v_2$, $\#v_3$) \rightarrow



Figure 2. Parametric flat automata of *x* and *y*

(0, 0, 0, 0, 1, 1, 501, 501).¹ The assignment says, for example, that *x* is the *parametric word* obtained by traversing the loop of A_x once (because $\#v_0 = \#v_1 = 1$), which is v_0v_1 . Under the assignment $v_0 = 0$ and $v_1 = 0$, we obtain x = "00".

If a model of the produced linear formula is found, then the procedure concludes SAT with an assignment to the string variables. If not, our procedure changes the PFAs to a more expressive one (by adding more states and transitions) and repeat the analysis. We report unknown after failing to prove SAT using a certain number of PFAs.

To demonstrate the usefulness of our approach, we have implemented our decision procedure in an open source solver, called Z3-Trau and evaluated it on a large set of benchmarks obtained from the literature and from symbolic execution of real world programs. The experimental results show that Z3-Trau is among the best tools for solving basic string constraints and significantly outperforms all other tools on benchmarks with string-number conversion constraints. In this benchmark, the total amount of tests cannot be solved by Z3-Trau is only a half to the second best tool.

Summary of the Contributions.

- An *efficient* procedure for checking satisfiability of string constraints with string-number conversion.
- The class of *parametric flat automata* which is the key for efficient handling of string constraints.
- An algorithm that translates the satisfiability problem of string constraints to the satisfiability problem of a linear formula in polynomial-time when the search space restricted by PFAs.
- An open source tool Z3-Trau with experimental results that demonstrate the efficiency of our approach on both existing and real-life benchmarks

Outline. After recalling the definition in Section 3, Section 4 presents a brief overview of our decision procedure. Section 5 introduces the class of parametric flat automata. Section 6 describes how to use PFA to restrict the searching domain of string variables. Section 7 shows how to construct the linear formula for basic string constraints (i.e., regular, equality, and integer constraints). Section 8 presents the construction of the linear formula for string-number conversion operations. Section 9 presents the details of our implementation and our experimental results. Related works are discussed in Section

10. Finally, Section 11 concludes the paper with a discussion of future works.

2 Preliminaries

We use \mathbb{N} and \mathbb{Z} to denote the sets of natural numbers and integers. For a set A, we use |A| to denote its size. For $n, m \in \mathbb{N}$, we write [n, m] for the set of natural numbers $\{k \mid n \leq k \leq m\}$. The function f with the domain restricted to a set D is denoted by f_D , and a set of functions F restricted to a set D is denoted by f_D , and a set of functions F restricted to a set D is $F_D = \{f_D \mid f \in F\}$. An *alphabet* is a finite set Σ of *characters* and a *word* over Σ is a sequence $w = a_1 \dots a_n$ of characters from Σ , with ϵ denoting the *empty word*. We use $w_1 \cdot w_2$ to denote the *concatenation* of words w_1 and w_2 . Σ^* is the set of all words over Σ , $\Sigma^+ = \Sigma^* \setminus \{\epsilon\}$ and $\Sigma_{\epsilon} = \Sigma \cup \{\epsilon\}$. A *language* over Σ is a subset L of Σ^* . We use |w| to denote the length of w and $|w|_a$ to denote the number of occurrences of the character $a \in \Sigma$ in w.

A finite automaton (FA) is a tuple (Q, T, Σ, q_i, q_f) , where Qis the set of states, $T \subseteq Q \times (\Sigma \cup \{\epsilon\}) \times Q$ is the set of transitions, Σ is the alphabet, q_i is the *initial state*, and q_f is the *final* state. A run π of A over a word $w = a_1 \cdots a_n$ is a sequence of transitions $(q_0, a_1, q_1), (q_1, a_1, q_2), \ldots, (q_{n-1}, a_n, q_n)$. The run π (resp. the word w) is accepting (resp. accepted) if $q_0 = q_i$ and $q_n = q_f$. The *language* of A (denoted by L(A)) consists of the set of all accepted words.

Through the paper, we will use quantifier-free linear integer arithmetic formulae, and call them *linear formulae* for short. Given a linear formula ϕ over variables V and an *integer interpretation* of V, a function $I : V \to \mathbb{Z}$, we denote by $I \models \phi$ that I satisfies ϕ (which is defined in the standard manner), and call I a *model* of ϕ . We use $\llbracket \phi \rrbracket$ to denote the set all models of ϕ .

The *Parikh image* of a word $w \in \Sigma^*$ maps each *Parikh variable #a*, where $a \in \Sigma$ is a character, to the number of occurrences of *a* in *w*. Formally, given a set *S*, let *#S* denote the set of Parikh variables $\{\#s \mid s \in S\}$. The Parikh image of *w* is a function $\mathbb{P}(w) : \#\Sigma \to \mathbb{N}$ such that $\mathbb{P}(w)(\#a) = |w|_a$ for each $a \in \Sigma$. The Parikh image of a language *L* is defined as follows $\mathbb{P}(L) = \{\mathbb{P}(w) \mid w \in L\}$. It is well known that the Parikh image of a regular language can be characterized by a linear formula.

Lemma 2.1 ([40]). Let A be a FA over the alphabet Σ . Then, we can compute, in linear time, a linear formula $\Phi_{\mathbb{P}}(A)$, over # Σ , such that $\llbracket \Phi_{\mathbb{P}}(A) \rrbracket_{\#\Sigma} = \mathbb{P}(L(A))$.

3 String Constraints

In this section, we formally define string constraints. To begin with, we fix a finite alphabet $\Sigma \subseteq \mathbb{N}$. Note that here we assume that the alphabet is a finite subset of natural numbers. Essentially, we try to capture the numerical encoding of the corresponding symbols in computers (e.g., in ASCII, 'A' is encoded as 65). Hence, we can assume w.l.o.g. that there is a one-to-one mapping between numbers in Σ and the

¹In these examples, we use the shorthand $(x_1, \ldots, x_k) \rightarrow (n_1, \ldots, n_k)$ to denote the function $\{x_1 \mapsto n_1, \ldots, x_k \mapsto n_k\}$.

character it encodes. For the simplicity of presentation, we assume that the character '0' is mapped to the number 0, '1' to 1,..., and '9' to 9. For other character c, we use [[c]] to denote the number that it maps to. Notice that this approach is general enough to support any finite set of characters.

A minor technical difficulty is that sometimes we may need to treat ϵ as a number. Therefore, we encode ϵ as some fixed number $\llbracket \epsilon \rrbracket \in \mathbb{N} \setminus \Sigma$.

Assume that X is a set of *string variables* ranging over Σ^* and Z a set of *integer variables* ranging over \mathbb{Z} . An *interpretation over* X *and* Z is a mapping $I : X \cup Z \to \Sigma^* \cup \mathbb{Z}$. A *word term* is an element in X^* . We lift the interpretation I to word terms and linear constraints in the standard manner.

We use four types of *atomic string constraint*:

- An equality constraint ϕ_e is of the form $t_1 = t_2$ where t_1, t_2 are word terms. The model of ϕ_e is the set of interpretations $[\![\phi_e]\!] = \{I \mid I(t_1) = I(t_2)\}$. A disequality constraint ϕ_d is of the form $t_1 \neq t_2$ and is interpreted analogously.
- An *integer constraint* ϕ_i is a linear constraint over the integer variables in *Z* and values of |x| for all $x \in X$, where $|\cdot| : X \to \mathbb{N}$ is the string length function defined in the standard way. We define $\llbracket \phi_i \rrbracket = \{I \mid I(\phi_i) = \text{true}\}.$
- A *regular constraint* φ_r is of the form x ∈ L(A) where x is a string variable and A is a finite automaton. The *model* of φ_m is the set of interpretations [[φ_m]] = {I | I(x) ∈ L(A)}.
- A string-number conversion constraint ϕ_s is of the form n = toNum(x), where the function toNum(x) is defined as follows. For $a \in [0, 9]$, we have toNum(a) = a and for $w \cdot a \in [0, 9]^+$, $\text{toNum}(w \cdot a) = 10 \times \text{toNum}(w) + a$. For $w \notin [0, 9]^+$, toNum(w) = -1. We define $[\![\phi_s]\!] = \{I \mid I(n) = \text{toNum}(I(x))\}$. The number-string conversion constraint x = toStr(n) is treated as a syntactic sugar for n = toNum(x). We assume decimal encoding of numbers.

A *string constraint* is then a conjunction of atomic string constraints, with the semantics defined in the standard manner. It is *satisfiable* if there is an interpretation which evaluates the constraint to true. Often we refer to the first three types of atomic string constraints the *basic string constraints*.

Notice that only positive integer is supported in the stringnumber conversion function. This is the semantics used by most of the SMT solvers, and hence we follow it in this paper. The encoding has a benefit that it can also handle the case where x is "not a number", using the condition toNum(x) = -1. Supporting only positive integer is not a strong restriction, since converting from negative integer can still be encoded using only the positive version.

4 Decision Procedure Overview

Our decision procedure has two steps: The first step consists in over-approximating the set of input constraints into a set that falls in the chain-free fragment [5], which is decidable. The over-approximation module proceeds as follows: First, it replaces all string-number conversion constraint n = toNum(x) by $n = -1 \lor (n \neq -1 \land x \in [0 - 9]^*)$ to obtain an over-approximation Φ consisting of only basic string constraints. Then, it over-approximates Φ to a *chain-free* string constraint [5], which consists of only integer, membership, and chain-free equality constraints. Informally, a set of equality constraints has a chain if we can find some circular dependency between the string variables in the equality constraints. Our procedure iteratively searches for such dependency chains in the equality constraints. If a chain is found then we replace a variable appearing in that chain by a fresh one. By doing this, we break that chain. We repeat this procedure until there are no more chains. Observe that if the over-approximation is UNSAT then our decision procedure declares that the original formula is also UNSAT.

The second step of our decision procedure is only enabled if the over-approximation step returns SAT. In this case, our decision procedure under-approximates the string constraints by restricting the search domain of each string variable to the language defined by some PFA. This approach based on PFA allows very efficient manipulation. We will show that given such restriction, one can reduce the string constraint solving problem to a linear formula satisfiability problem. The rest of the paper will be mainly dedicated to the explanation of the under-approximation technique (which is our main contribution).

5 Parametric Flat Automata

We introduce *parametric flat automata* that will be used to define patterns used by the under-approximation module to restrict the domain of string variables.



Flat Automata. A finite state automaton $A = (Q, T, \Sigma, q_0^0, q_m^0)$ is said to be *flat* if it satisfies the following structural constraints (see also the figure above):

- 1. The final state q_m^0 is reached from the initial state q_0^0 through a straight path of m 1 transitions $(q_i^0, a_i, q_{i+1}^0) \in T$ with $q_i^0 \in Q$ and $a_i \in \Sigma$ for $i \in [0, m 1]$.
- 2. Each state q_i^0 is the origin of a unique simple cycle of the length $l_i \in \mathbb{N}$, consisting of states $q_i^{j-1} \in Q$ and transitions $(q_i^{j-1}, a_i^{j-1}, q_i^{j \mod l_i})$, with $a_i^{j-1} \in \Sigma$, for $j \in [1, l_i]$. Notice that the case when $l_i = 0$ is also admissible and means that there is no cycle on q_i .

 Each character in Σ appears on at most one transition of the automaton A.

The crucial feature of flat automata is that their semantics can be faithfully represented by a linear formula and handled efficiently by an SMT solver. Such encoding into linear formula results in efficient algorithms and decision procedures. For instance, we avoid dealing with costly standard automata operations (e.g., checking the non-emptiness of the intersection of several regular languages is known to be PSPACE-complete while it is in NP for the class of flat automata). The encoding is possible due to the flat structure, which has the property that "every word $w \in L(A)$ is uniquely determined by its Parikh image $\mathbb{P}(w)$ ". More precisely, the Parikh image of a word $w \in L(A)$ can be seen as an encoding of w and can be uniquely decoded:

Lemma 5.1. For a flat FA A, there is a function decode_A such that for each $w \in L(A)$, decode_A($\mathbb{P}(w)$) = w.

Observe that the $\mathbb{P}(w)$ value of any variable appearing within a cycle of *A* is equal to the number of repetitions of that cycle in the accepting run. This is an immediate consequence of the fact that every character appears on at most one transition. Thus, the accepting run on *w* (and so *w* itself) can be reconstructed from $\mathbb{P}(w)$.

More concretely, the function $decode_A$ can be implemented as follows. Given $I_{\#} : \#\Sigma \to \mathbb{N}$, and assuming that the lengths of the loops of A are l_0, \ldots, l_m , $decode_A(I_{\#})$ is constructed as the word $w_0 a_0 w_1 \cdots a_{m-1} w_m$ where for each $i \in [0, m]$, $w_i = (a_i^0 \cdots a_i^{l_i-1})^{\#a_i^0}$ if $l_i > 0$ and $w_i = \epsilon$ if $l_i = 0$.

For example, in the automaton given at the beginning of this section, from $|x|_{a_0} = |x|_{a_1} = \cdots = |x|_{a_{m-1}} = 1$, $|x|_{a_1^0} = |x|_{a_1^1} = |x|_{a_1^2} = 2$ and $|x|_{a_j^i} = 0$ otherwise, we derive that $x = a_0(a_1^0a_1^1a_1^2)^2a_1\cdots a_{m-1}$.

Parametric (Flat) Automata. Next, we define *parametric automaton* (PA) as a pair $P = (A, \psi)$ where A is an automaton operating over an alphabet V of *character variables* and ψ is an *interpretation constraint*, a linear formula over V. *Parametric flat automaton* (*PFA*) is then a parametric automaton whose automaton is flat. See Figure 1 and 2 for examples of PFAs (without interpretation constraints, i.e., ψ = true.).

Parametric automata accept words over V, called *parametric words*, but we still use them as representations of languages over Σ . Namely, words over V are interpreted as words over Σ under an *interpretation of* V, a mapping $I : V \rightarrow \Sigma_{\epsilon}$ (recall that $\Sigma_{\epsilon} ::= \Sigma \cup {\epsilon} \subseteq \mathbb{N}$). For a parametric word $x = v_1v_2\cdots v_k$ over V, its interpretation I(x) is then defined as $I(v_1) \cdot I(v_2) \cdot \ldots \cdot I(v_k)$. We then define the semantics of the PA P as the set of strings $[\![P]\!] = {I(x) \mid x \in L(A), I \in [\![\psi]\!]}$ of all interpretations satisfying ψ of all parametric strings in the language of A.

We say that a mapping $I_e : V \cup \#V \rightarrow \mathbb{N}$ is a *word encoding* of a word *w* (or a *P*-encoding of *w*) if *w* is an instantiation of some parametric word $x \in L(P)$ whose Parikh image

and interpretation of character variables are defined by I_e . Conversely, w is a *P*-decoding of I_e . We use $encode_P(w)$ below to denote all *P*-encodings of a word w, and $decode_P(I_e)$ to denote all *P*-decodings of a word encoding I_e . Namely,

 $encode_P(w) = \{I_e | x \in L(P), I(x) = w, I \in \llbracket \psi \rrbracket, I_e = I \cup \mathbb{P}(x)\}$ $decode_P(I_e) = \{w | x \in L(P), I(x) = w, I \in \llbracket \psi \rrbracket, I_e = I \cup \mathbb{P}(x)\}$

Since a word encoding I_e only records the numbers of occurrences of character variables (Parikh image), the same word encoding may be shared by multiple words, as formalized in the definition of $decode_P(I_e)$.

Example 5.2. Let use consider the PFA $P_x = (A_x, \text{true})$ from Figure 2 (a) and let $Y = (v_1, v_2, \#v_1, \#v_2)$. Then we have $encode_{P_x}(\text{``aaa''}) = \{(Y \rightarrow (\llbracket a \rrbracket, \llbracket \epsilon \rrbracket, 3, 3), Y \rightarrow (\llbracket \epsilon \rrbracket, [\llbracket a \rrbracket, 3, 3)\}$ and $decode_{P_x}((Y \rightarrow (\llbracket a \rrbracket, \llbracket \epsilon \rrbracket, 3, 3)) = \{\text{``aaa''}\}.$

If *P* is a PFA, then by Lemma 5.1, every word encoding $I_e \in encode_P(w)$ can be decoded uniquely to the word *w*, i.e.

 $\{w\} = decode_P(encode_P(w))$

Similarly, as stated by the following corollary of Lemma 5.1, Parikh image of parametric words in L(A) paired with character variable interpretations satisfying ψ encode precisely the words in $\llbracket P \rrbracket$.

Corollary 5.3. For a PFA $P = (A, \psi)$, $[\![P]\!] = decode_P(\{(I \cup I_{\#}) \mid I_{\#} \in \mathbb{P}(L(A)), I \in [\![\psi]\!]\}).$

6 Flat Domain Restriction

In this section, we describe formally how to restrict the domain of string variables to patterns defined by PFA. We start the description of the algorithm that converts a string constraint ϕ_{in} to a linear formula representing the set of solutions under the domain restriction.

The domain restriction is formally defined by restricting the domain of each string variable by a chosen PFA. Namely, assuming that X is the set of string variables of ϕ_{in} , a *flat domain restriction* for ϕ_{in} is a mapping \mathcal{R} that assigns to each variable $x \in X$, a PFA $\mathcal{R}(x)$ over character variables V_x . Let $V_{\mathcal{R}} = \bigcup_{x \in X} V_x$ be the set of all character variables used in \mathcal{R} . We require that these PFA operate over pairwise disjoint sets of character variables, that is if $x \neq y$ then $V_x \cap V_y = \emptyset$. The particular choice of a PFA for each variable depends on the strategy used in the implementation, and will be discussed in Section 9. The *flattening* of the input string constraint ϕ_{in} , denoted $flatten_{\mathcal{R}}(\phi_{in})$, will be built inductively following the structure of ϕ_{in} . For a conjunction of string constraints, we let $flatten_{\mathcal{R}}(\phi \land \phi') ::= flatten_{\mathcal{R}}(\phi) \land flatten_{\mathcal{R}}(\phi')$. We do such decomposition until reached atomic string constraints. We show how to build a flattening *flatten*_{\mathcal{R}}(ϕ) for every atomic string constraint ϕ in the following sections.

The semantics of a string constraint ϕ restricted by \mathcal{R} is then defined as $\llbracket \phi \rrbracket^{\mathcal{R}} = \{I \in \llbracket \phi \rrbracket \mid \forall x \in X : I(x) \in \llbracket \mathcal{R}(x) \rrbracket\}.$

The correctness of the entire construction of $flatten_{\mathcal{R}}(\phi_{in})$ is expressed by Theorem 6.2. It uses the decoding function

 $decode_{\mathcal{R}}$ parameterized by the domain restriction \mathcal{R} . Let Z be the set of integer variables in ϕ_{in} . The function maps an interpretations I_e over $Z \cup V_R \cup \#V_R$ to an interpretation over $Z \cup X$, following the domain restriction \mathcal{R} . Informally, it "decodes" an interpretation of integer variables $Z \cup V_{\mathcal{R}} \cup #V_{\mathcal{R}}$ to an interpretation of variables in the string constraint ϕ_{in} . Formally, we define $decode_{\mathcal{R}}(I_e) ::= \{I \mid \forall z \in Z : I(z) = I_e(z) \land \forall x \in I_e(z) \land x \in X$ $X : \{I(x)\} = decode_{\mathcal{R}(x)}((I_e)_{V_{\mathcal{R}(x)} \cup \#V_{\mathcal{R}(x)}})\}$. The condition says that (1) I and I_e are consistent over variables in Z and (2) $(I_e)_{V_{\mathcal{R}(x)} \cup \#V_{\mathcal{R}(x)}}$ is a word encoding that $\mathcal{R}(x)$ -encodes I(x). We also define the R-encoding function as the counterpart of \mathcal{R} -decoding, namely, for a interpretation I of the string constraint ϕ_{in} , we let $encode_{\mathcal{R}}(I) = \{I_e \mid decode_{\mathcal{R}}(I_e) = \{I\}\}$. We lift $decode_{\mathcal{R}}$ and $encode_{\mathcal{R}}$ to sets of interpretations in the standard manner.

Example 6.1. We consider the domain restriction \mathcal{R} such that $\mathcal{R}(x) = (A_x, \text{true})$ from Figure 2 (a) and $\mathcal{R}(y) =$ $(A_y, true)$ from Figure 2 (b). Let the set of integer variables be $Z = \{v_z\}$ and let $V_{\mathcal{R}(x)} = \{v_0, v_1, v_2, v_3\}$. For the interpretation $I_e = (v_z, v_0, v_1, v_2, v_3, \#v_0, \#v_1, \#v_2, \#v_3) \rightarrow$ $(3, [[a]], [[b]], [[c]], [[\epsilon]], 3, 3, 2, 2)$, we have $decode_{\mathcal{R}}(I_e)$ = $\{(z, x, y) \rightarrow (3, ``ababab", ``cc")\}.$

Conversely, for $I = (z, x, y) \rightarrow (3, "ababab", "cc")$ and Y = $(v_z, v_0, v_1, v_2, v_3, \#v_0, \#v_1, \#v_2, \#v_3)$, we have $encode_{\mathcal{R}}(I) =$ $\begin{array}{l} Y \to (3, [[a]], [[b]], [[c]], [[e]], 3, 3, 2, 2), \\ Y \to (3, [[a]], [[b]], [[e]], [[e]], 3, 3, 2, 2), \\ Y \to (3, [[a]], [[b]], [[e]], [[c]], 3, 3, 1, 1) \end{array}$

Theorem 6.2. $decode_{\mathcal{R}}(\llbracket flatten_{\mathcal{R}}(\phi_{in}) \rrbracket) = \llbracket \phi_{in} \rrbracket^{\mathcal{R}}$

The theorem can be proved by a structural induction over ϕ_{in} . However, for the induction step to go through, we will need to guarantee a stronger correspondence of string constraints ϕ and their flattenings *flatten*_R(ϕ) than just the semantic equality $decode_{\mathcal{R}}(\llbracket flatten_{\mathcal{R}}(\phi) \rrbracket) = \llbracket \phi \rrbracket^{\mathcal{R}}$. Particularly, we will need to ensure that $\mathit{flatten}_{\mathcal{R}}(\phi)$ captures exactly all \mathcal{R} -encodings of $\llbracket \phi \rrbracket^{\mathcal{R}}$ (indeed, notice that if it would be allowed to capture only some of the encodings, then for instance $flatten_{\mathcal{R}}(\phi) \wedge flatten_{\mathcal{R}}(\phi')$ could only underapproximate $\llbracket \phi \land \phi' \rrbracket^{\mathcal{R}}$). The inductive argument needed in the correctness proof of the under-approximation then reads as

$$\llbracket flatten_{\mathcal{R}}(\phi) \rrbracket_{V_{\mathcal{R}} \cup V_{\#\mathcal{R}}} = encode_{\mathcal{R}}\llbracket \phi \rrbracket$$

In the next sections, we will formulate the corresponding correctness lemma for flattening constructed from each type of atomic string constraints. We note that the restriction of [[*flatten*_{\mathcal{R}}(ϕ)]] to $V_{\mathcal{R}} \cup V_{\#\mathcal{R}}$ here is needed since *flatten*_{\mathcal{R}}(ϕ) will be constructed with some auxiliary variables.

Flattening of Basic String Constraints 7

We will first discuss flattening of the basic string constraints, that is, regular, equality, and integer constraints. We start by two needed operations over PA, synchronization and concatenation.

Synchronization of PAs. We will now discuss a construction of the synchronization formula for two PAs P and P'. It is a linear formula $\Psi_{P \times P'}$ that specifies how each word in the semantic intersection $\llbracket P \rrbracket \cap \llbracket P' \rrbracket$ is encoded by *P* and by *P'*. More precisely, the models of $\Psi_{P \times P'}$ represent pairs of word encodings I_e and I'_e such that $I_e \in encode_P(w)$ and $I'_e \in encode_{P'}(w)$ (hence $w \in \llbracket P \rrbracket \cap \llbracket P' \rrbracket$).

Particularly, the synchronization formula is built for two PAs $P = ((Q, T, V, q_i, q_f), \psi)$ and $P' = ((Q', T', V', q'_i, q'_f), \psi')$ such that $V \cap V' = \emptyset$. It is extracted from the asynchronous product of P and P'. The asynchronous product is an automaton that uses $Q \times Q'$ as the set of states and $V_{\epsilon} \times V_{\epsilon}'$ as the alphabet. Every accepting run of $P \times P'$ corresponds to a pair of accepting runs, a run of P over a parametric word xand a run of P' over a parametric word x'. The word accepted by the run of $P \times P'$ induces constraints on the interpretations of I over V and I' over V' under which the two parametric words have the same interpretation, i.e. I(x) = I'(x').

Intuitively, when the product automaton $P \times P'$ takes a transition $((q_1, q'_1), (v, v'), (q_2, q'_2))$, it means the character variable v and v' should be assigned the same value, P moves under v from state q_1 to state q_2 and P' from q'_1 to q'_2 under v'. When $P \times P'$ takes a transition $((q_1, q'_1), (v, \epsilon), (q_2, q'_1))$, it means that the character variable v should be assigned ϵ , *P* moves under v to q_1 , and *P'* takes no action, since no action is needed to match *P*'s reading of ϵ (hence consumes no symbol from the input word). Symmetrically, P' might read a variable v assigned ϵ and P may stay.

Formally, the asynchronous product automaton is a tuple $P \times P' = (Q \times Q', T_{\times}, V_{\epsilon} \times V'_{\epsilon}, (q_i, q'_i), (q_f, q'_f))$, where the transition relation T_{\times} is the minimal set satisfying the following:

- If $(q_1, v, q_2) \in T$ and $(q'_1, v', q'_2) \in T'$, then we have $((q_1, q'_1), (v, v'), (q_2, q'_2)) \in T_{\times}.$
- If $(q_1, v, q_2) \in T$, then for all states $q' \in Q'$, we have $((q_1, q'), (v, \epsilon), (q_2, q')) \in T_{\times}.$
- If $(q'_1, v', q'_2) \in T'$, then for all states $q \in Q$, we have $((q, q'_1), (\epsilon, \upsilon'), (q, q'_2)) \in T_{\times}.$

The synchronization formula $\Psi_{P \times P'}$ is extracted from $P \times P'$ as follows. Its first part is the Parikh formula $\Phi_{\mathbb{P}}(P \times P')$ of the product, which encodes all accepting runs of $P \times P'$. The second part is a constraint that extracts from a run of $P \times P'$ the corresponding runs of *P* and of *P*':

$$\Psi_{\#} ::= \left(\bigwedge_{\upsilon \in V} \#\upsilon = \sum_{x' \in V'_{\epsilon}} \#(\upsilon, x') \right) \land \left(\bigwedge_{\upsilon' \in V'} \#\upsilon' = \sum_{x \in V_{\epsilon}} \#(x, \upsilon') \right)$$

Notice that x, x' are either variables or ϵ . Finally, the third part forces the interpretations of the parametric words accepted by P and P' to be the same:

$$\Psi_{=} ::= \bigwedge_{x \in V_{\epsilon}, x' \in V_{\epsilon}'} \#(x, x') > 0 \to (x = x')$$

The synchronization formula is then the conjunction

$$\Psi_{P\times P'} ::= \Phi_{\mathbb{P}}(P\times P') \land \Psi_{\#} \land \Psi_{=} \land \psi \land \psi'$$

The correctness of this construction is stated in Lemma 7.1 below. The correctness of the construction of underapproximations of equality constraints and regular constraints in Section 7.1 and Section 7.2 rely on it.

Lemma 7.1.
$$\llbracket \Psi_{P \times P'} \rrbracket_{V \cup V' \cup \#V \cup \#V'} = \{I_e \cup I'_e \mid I_e \in encode_P(w), I'_e \in encode_{P'}(w), w \in \llbracket P \rrbracket \cap \llbracket P' \rrbracket \}$$

Informally, the lemma states that the models of $\Psi_{P \times P'}$ encode precisely the pairs of equivalent encodings of words from $\llbracket P \rrbracket$ and $\llbracket P' \rrbracket$, that constitute the intersection $\llbracket P \rrbracket \cap \llbracket P' \rrbracket$. Since that the models of $\Psi_{P \times P'}$ include also an assignment to the auxiliary variables of $(V_{\epsilon} \times V'_{\epsilon}) \cup \#(V_{\epsilon} \times V'_{\epsilon})$, the lemma restricts $\llbracket \Psi_{P \times P'} \rrbracket$ to $V \cup V' \cup \#V \cup \#V'$.

Notice that if *P* is flat (or, symmetrically, if *P'* is flat), then the semantic intersection $\llbracket P \rrbracket \cap \llbracket P' \rrbracket$ can be still decoded from the synchronization formula $\Psi_{P \times P'}$. Namely, due to Corollary 5.3, we have that if *P* is a PFA, then

$$decode_P(\llbracket \Psi_{P \times P'} \rrbracket_{V \cup \#V}) = \llbracket P \rrbracket \cap \llbracket P' \rrbracket$$

Concatenation of PFAs. Concatenation of PFAs will be needed when flattening equality constraints. Its implementation is straightforward, connect the final state of the first PFA with the initial state of the second by an ϵ -transition. Since our automata do not allow transition directly labeled by ϵ , the ϵ -transition is labeled by a fresh variable v_{ϵ} forced by the constraint $v_{\epsilon} = \epsilon$ to take the value ϵ .

Formally, given PFA $P = ((Q, T, V, q_i, q_f), \psi)$ and $P' = ((Q', T', V', q'_i, q'_f), \psi')$ with $Q \cap Q' = \emptyset = V \cap V'$, their concatenation is the PFA $P \cdot P' = (Q \cup Q', T \cup T' \cup \{(q_f, v_{\epsilon}, q'_i)\}, V \cup V' \cup \{v_{\epsilon}\}, q_i, q'_f, \psi \land \psi' \land v_{\epsilon} = \epsilon)$ where v_{ϵ} is fresh, not from $V \cup V'$.

Lemma 7.2. $encode_{P\cdot P'}(\llbracket P \cdot P \rrbracket)_{V \cup V' \cup \#V \cup \#V'} = \{I_e \cup I'_e \mid I_e \in encode_P(\llbracket P \rrbracket) \land I'_e \in encode_{P'}(\llbracket P' \rrbracket)\}, for PFAs P and P'.$

With synchronization and concatenation of PA, we are ready to describe flattening of the basic string constraints.

7.1 Flattening of Regular Constraints

Let us first describe the construction of $flatten_{\mathcal{R}}(\phi_r)$ for a regular constraint $\phi_r ::= x \in L(A)$. In order to synchronize the FA A with the PFA $\mathcal{R}(x)$, we represent A by a PA $P' = (A', \Psi_{char})$. The automaton A' of P' operates over fresh character variables $v_a, a \in \Sigma_{\epsilon}$, and is obtained from A by replacing every occurrence of each character $a \in \Sigma_{\epsilon}$ on a transition by the variable v_a . The interpretation restriction formula Ψ_{char} of P' then binds the fresh character variables to the character values they represent, namely, $\Psi_{char} = \bigwedge_{a \in \Sigma_{\epsilon}} v_a = [\![a]\!]$. Obviously, $L(A) = [\![P']\!]$. We then let

$$flatten_{\mathcal{R}}(\phi_r) = \Psi_{\mathcal{R}(x) \times P'}$$

The following lemma states the correctness of this construction. It follows from Corollary 5.3 and Lemma 7.1. **Lemma 7.3.** $\llbracket flatten_{\mathcal{R}}(\phi_r) \rrbracket_{V_{\mathcal{R}} \cup \#V_{\mathcal{R}}} = encode_{\mathcal{R}}(\llbracket \phi_r \rrbracket)$.

7.2 Flattening of Equality Constraints

We now describe the construction of $flatten_{\mathcal{R}}(\phi_e)$ for an equality constraint $\phi_e ::= x_0 \cdot x_1 \cdots x_n = x_{n+1} \cdot x_{n+2} \cdots x_m$. To simplify the presentation, we assume that the variables are pairwise different, i.e. that $i \neq j \implies x_i \neq x_j$. We may make this assumption without loss of generality, since multiple occurrences of variables in ϕ_e can be eliminated. In fact, whenever $x_i = x_j$ for $i \neq j$, we may replace x_j by a fresh variable x'_j and conjoin the modified ϕ_e with a new equality $x_j = x'_j$. We also assume that all disequality constraint $t \neq t'$ are already converted to equivalent equality constraints and integer constraints in the standard way [4].

Having made these assumptions, we may proceed follows. First, we build two PFAs P^{left} and P^{right} that encode the left and the right-hand side word term of the equality constraint ϕ_e , respectively, by concatenating the restrictions of the individual variables. That is

$$P^{left} ::= \mathcal{R}(x_1) \cdot \ldots \cdot \mathcal{R}(x_n) \quad P^{right} ::= \mathcal{R}(x_{n+1}) \cdot \ldots \cdot \mathcal{R}(x_m)$$

The under-approximation of ϕ_e is then obtained as their synchronization formula

$$flatten_{\mathcal{R}}(\phi_e) = \Psi_{P^{left} \times P^{right}}$$

The correctness of the construction is stated by the lemma:

Lemma 7.4. $\llbracket flatten_{\mathcal{R}}(\phi_e) \rrbracket_{V_{\mathcal{R}} \cup \#V_{\mathcal{R}}} = encode_{\mathcal{R}}(\llbracket \phi_e \rrbracket).$

7.3 Flattening of Integer Constraints

Given an integer constraint ϕ_l that talks about lengths |x| of string variables $x \in X$. We use a version of ϕ_l where every occurrence of every length function |x| is replaced by an auxiliary length variable l_x and we add a formula to ensure that the value of l_x is equal to that of |x| even when x is encoded using the character variables V_x and Parikh variables $\#V_x$ of $\mathcal{R}(x)$. We will need a set auxiliary variables $\{l_v \mid v \in V_x\}$ to express the length by which the character variable v contributes to the length of an \mathcal{R} -encoded string x. That is, l_v will be 0 if v is assigned $[\![\epsilon]\!]$, otherwise it will equal to the number #v of its occurrences in the word:

$$\Psi_{l_{v}} ::= (v = \llbracket \epsilon \rrbracket \land l_{v} = 0) \lor (v \neq \llbracket \epsilon \rrbracket \land l_{v} = \# v)$$

The length of the encoded word x is then the sum of the lengths contributed by the individual character variables in V_x , hence we let

$$\Psi_{l_x} ::= l_x = \sum_{\upsilon \in V_x} l_\upsilon \wedge \bigwedge_{\upsilon \in V_x} \Psi_{l_\upsilon} \ .$$

Finally, the linear formula created for ϕ_l is

$$flatten_{\mathcal{R}}(\phi_l) ::= \phi_l \wedge \bigwedge_{x \in X} \Psi_{l_x}$$

and the following lemma states its correctness.

Lemma 7.5. $[[flatten_{\mathcal{R}}(\phi_l)]]_{V_{\mathcal{R}} \cup \#V_{\mathcal{R}}} = encode_{\mathcal{R}}([[\phi_l]])$

8 Flattening of String-Number Conversion

Last, we present the main contribution of this paper, the construction of a flattening $flatten_{\mathcal{R}}(\phi_s)$ of string-number conversion constraint $\phi_s ::= n = \text{toNum}(x)$.

Let us begin with a simple example. Assume that we use the PFA in Figure 2 (a) to restrict the domain of *x*. Then we know that when $0 \le v_0, v_1 \le 9$, then *n* is a positive integer value, and otherwise n = -1. So we should first add the constraint $((0 \le v_0 \le 9) \land (0 \le v_1 \le 9)) \lor (n = -1 \land (v_0 > 9 \lor v_1 > 9))$.

For the case that *n* is a positive integer, the value of *n* can be characterized by a constraint (assume character variables are not assigned ϵ) $n = (v_0 \times 10 + v_1) \times (1 + 100 + 100^2 + ... 100^{\#v_0-1}) = (v_0 \times 10 + v_1) \times \frac{100^{\#v_0-1}}{100-1}$. The constraint uses $\#v_0$ to capture the total number of loop traversals. Notice that the constraint above contains an exponential component $\frac{100^{\#v_0}}{100-1}$. To solve the satisfiability of this formula, one needs to solve an exponential constraint.

Let us have a look at another example. If we restrict the domain of *x* to the PFA in Figure 2 (c), for the case when *n* is positive, we have the relation $n = (v_0 \times 10 + v_1) \times (1 + 100 + 100^2 + ... 100^{#v_0-1}) \times 10 \times 100^{#v_3} + v_2 \times 100^{#v_3} + (v_3 \times 10 + v_4) \times (1 + 100 + 100^2 + ... 100^{#v_3-1}) = (v_0 \times 10 + v_1) \times \frac{100^{#v_0-1}}{100-1} \times 10 \times 100^{#v_3} + v_2 \times 100^{#v_3} + (v_3 \times 10 + v_4) \times \frac{100^{#v_3-1}}{100-1}$. Observe that the formula has multiple exponential components, including $\frac{100^{#v_3} \times 100^{#v_3}}{100-1}$ and $\frac{100^{#v_3}}{100-1}$.

It is not difficult to see from the examples above that, if $\mathcal{R}(x)$ is an arbitrary PFA with *m* loops, the formula that defines the number *n* contains at least *m* exponential components, one for each loop. To the best of our knowledge, the satisfiability problem of integer constraints with a mix of polynomials and exponentials is still open. The problem is difficult even for the case that variables are real numbers [17, 18]. For example, the algorithm in [24] involves a quantifier elimination procedure which is double-exponential to the length of the input formula and hence cannot handle large instances. We therefore do not expect that such constraint can be solved efficiently. Instead, we will define a special form of the flat restriction $\mathcal{R}(x)$ of *x* that leads to an easier linear formula.

Efficient String-Number Conversion using PFA. We will now discuss the special form of $\mathcal{R}(x)$, called *numeric PFA*, which we choose for string variables appearing in string-integer constraints and that leads to efficient under-approximation technique. Besides simple induced linear formulae, we still want single $\mathcal{R}(x)$ to cover as many numerals as possible. We want an "easy" completeness property, that is, (1) the space of all numerals can be covered completely by our special numeric PFAs, (2) these numeric PFAs are generated easily, and (3) each of them covers a large and practically significant portion of numerals (so that satisfiable assignments can be often found within the domain restriction of



Figure 3. PFA for string-number conversion constraints

just few of numeric PFAs). Particularly, we will proceed towards a definition of *numeric PFA* $(A^m, \psi^m), m \in \mathbb{N}$ which covers all numerals that encode integers with *m* digits.

Our first attempt is a PFA without loop, i.e., a straight line structure $\xrightarrow{q_0} v_1 \xrightarrow{q_1} v_2 \xrightarrow{\cdots} v_m \xrightarrow{q_m} q_m$. The corresponding integer constraint does not have exponential components. However, it does cover all numbers with at most *m*-digits. Consider the example toNum(*x*) = 10 \land |*x*| = 5. The number 10 has only 2-digits, at the first glance, a straight-line PFA with two transitions, i.e., the PFA $\xrightarrow{-q_0} v_1 \xrightarrow{q_1} v_2 \xrightarrow{q_2}$ should be sufficient for the domain restriction of *x*. If we do so, we will conclude that the formula is unsatisfiable, because the length of *x* cannot be 5 under this domain restriction.

However, the formula is satisfiable when x = "00010". Observe that toNum("00010") = 10 since PFAs accept numerals with the least significant bit first. The key is that even for a bounded integer, the corresponding numeral can be of an unbounded length with arbitrarily many trailing '0's at the front. All numbers with up to *m*-digits can be however still handled without having to solve exponential constraints. It is enough to equip the initial state of the PFA with a 0-self-loop.

Consequently, the automaton A^m of our numeric PFA will have the following form illustrated in Figure 3. It has a selfloop on the initial state labeled by the character variable v_0 , forced by the constraint

$$\Psi_{\upsilon_0} ::= \upsilon_0 = 0$$

to hold the value 0. This transition ensures that the underapproximation handles numerals with arbitrary number of trailing zeros. The self-loop is followed by a chain of *m* transitions $(q_{i-1}, v_i, q_i), 1 \le i \le m$, leading towards the final state q_m . The chain encodes at most *m* meaningful digits (only *at most* because the first variables in the sequence can still be assigned zeros and some variables may be assigned ϵ). Hence this PFA covers all numerals that encode numbers with at most *m* digits. Although it still has a loop, it will not create any exponential component defining the value of *n* because the loop only represents a sequence of "0" at the front of *x*. Thus, it will not affect the integer value of n = toNum(x).

Numeric PFA with these restrictions would satisfy our primary objective, that is, they would induce linear formulae and would "easily" and completely cover all numerals. A last problem still needs to be solved before they can be efficient in practice. Recall that the character variables can be assigned ϵ . Therefore, a single chain of k interesting digits, $k \leq m$, can be by A^m represented in $\binom{k}{m}$ ways, each corresponding to one possible interleaving of k digits with m-k epsilons. This may

lead to a formula of exponential size when defining the value of *n*. In order to eliminate this potential blow-up in the size of the formula, we add to ψ^m an additional constraint that forces all epsilons to be *shifted* behind the least significant digit. This will leave us with only one interleaving. This is the formula

$$\Psi^m_{\text{shift}} ::= \bigwedge_{1 \le i \le m} v_i \neq \llbracket \epsilon \rrbracket \implies v_{i-1} \neq \llbracket \epsilon \rrbracket.$$

Last, since this restriction is meaningful only when the string is indeed a numeral, we also define the constraint representing the strings which are not numerals, the formula

$$\Psi_{\mathsf{NaN}}^m ::= \bigvee_{i \in [1,m]} v_i > 9$$

and define the final form of the interpretation restriction used by A^m as

$$\psi^m ::= \Psi^m_{\mathsf{NaN}} \lor (\Psi_{\upsilon_0} \land \Psi^m_{\mathsf{shift}})$$

Consequently, we design our domain restrictions \mathcal{R} so that for string variables x that appear within string-integer constraints, $\mathcal{R}(x)$ is a numeric PFA $(A^m, \psi^m), m \in \mathbb{N}$.

Assuming that the domain restriction for *x* is (A^m, ψ^m) , the value of the integer *n* can be extracted from a numeral using the formula

$$\Psi_{\text{toInt}}^{m} ::= \bigvee_{1 \le k \le m} \Psi_{\text{last}(k)}^{m} \land (n = v_1 * 10^{k-1} + v_2 * 10^{k-2} + \ldots + v_k)$$

where $\Psi_{\text{last}(k)}^m$ says that the last variable of A^m assigned a non- ϵ value is v_k , namely

$$\Psi_{\mathsf{last}(k)}^{m} ::= (k = m \land v_{k} \ge 0) \lor (v_{k} \ge 0 \land v_{k+1} = -1)$$

Since we also need to distinguish the case when x is not a number, in which case n should equal -1, the formula under-approximating ϕ_s is finally constructed as

$$flatten_{\mathcal{R}}(\phi_{s}) ::= \Phi_{\mathbb{P}}(A^{m}) \land$$
$$\left((\Psi_{\text{NaN}}^{m} \land n = -1) \lor (\neg \Psi_{\text{NaN}}^{m} \land \Psi_{\text{tolnt}}^{m}) \right)$$

The following lemma states correctness of this construction:

Lemma 8.1. $\llbracket flatten_{\mathcal{R}}(\psi_s) \rrbracket_{V_{\mathcal{R}} \cup \#V_{\mathcal{R}}} = encode_{\mathcal{R}}(\llbracket \phi_s \rrbracket)$

9 Implementation and Evaluation

We have implemented our string constraint solving procedure in a tool called Z3-Trau. Z3-Trau is implemented as a theory solver of the SMT solver Z3 [15]. In this way, we can concentrate on solving conjunctive constraints and let Z3 handle the other boolean connectives. Secondly, it makes it possible to solve not only formulae over string constraints but also combinations of string constraints with other theories that Z3 supports. Furthermore, this approach allows us to more effectively handle the arithmetic constraints that are generated by the under-approximation module and, lastly, it eliminates the need to have our own parser. In Z3-Trau, we use the following PFA selection strategy. We use *numeric PFAs* for string variables appearing in stringnumber conversion and *standard PFAs* for others. We select a size *m* for numeric PFAs, a number *p* of their loops, and the length *q* of the loops. Initially, we set (m, p, q) = (5, 2, q)where *q* is dynamic and obtained from our internal static analysis. We double *m* and increase *p* and *q* by one if refinement is required. We set an upper bound for each parameter and report UNKNOWN if a solution cannot be found within the bound.

Our over-approximation module also uses heuristics to derive the constant value of any side of the constraint n = toNum(x) to refine the over-approximation. For instance, assume we can derive that n = 12 from some integer constraints. Then we can derive the value of x belongs to the regular language (0^{*}12).

The way our theory solver and Z3 interact is almost standard. When Z3 asks our theory solver a string constraint satisfiability problem, our solver tries to prove it is SAT or UNSAT using the procedure discussed in this paper. For under-approximation, whenever a corresponding linear formula is created, we attach the current value of m, p, q to the formula, and then push it to Z3 core. If our theory solver reports UNKNOWN, Z3 remembers it in a global flag incomplete and either tries another solution branch, or the same solution branch with different value of m, p, q. If Z3 completes the search of all solution branches, it reports UNSAT if the flag incomplete is down, and UNKNOWN otherwise.

We compare Z3-Trau (1e715b7dab)² with other state-ofthe-art string solvers, namely, CVC4 (version 1.7) [8], Z3 (version 4.8.7) [15], and Z3Str3 (version 4.8.7) [47]. For these tools, the versions we used are the latest release version. Observe that CVC4 and Z3 are DPLL(T)-based string solvers. We do not compare with Sloth [21] since it does not support length constraints, which occur in most of our benchmarks. We also do not compare with ABC [6] (a model counter for string constraints), Ostrich [13] and Trau+ [5], because they do not support many of the string functions in our benchmarks, especially string-number conversion.

We perform two sets of experiments. In the first set of experiments, we compare Z3-Trau with other tools on existing benchmarks over basic string constraints. Those benchmarks do not involve string-number conversion functions. In the second set of experiments, we compare Z3-Trau with the other tools on new suites focusing on string-number conversion. Our goals of experiments are the following:

- Z3-Trau performs as good as or better than the other tools in solving the satisfiability problems of basic string constraints.
- Z3-Trau performs significantly better than the other tools in solving the satisfiability problems on string-number conversion benchmark, and this shows the

```
<sup>2</sup>https://github.com/guluchen/z3/tree/1e715b7dab
```

efficiency of PFA in general and *numeric* PFA in particular.

In the first set of experiments, we use the following benchmark examples:

- PyEx [34] comes from running the symbolic executor PyEx over some Python packages.
- LeetCode comes from running PyEx over a sample code collected from the LeetCode [25] website, including functions that check whether a string is a valid IPv4 or IPv6 address, sum up two binary numbers, check whether an input string is an abbreviation of another input string, and convert a sequence of digits to a string according to a given mapping.
- StringFuzz [10] is generated by the fuzz testing tool of the same name.
- cvc4_{pred} and cvc4_{term} are obtained from the CVC4 group [33]. These benchmarks contain a small amount of string-number conversion constraints (< 5%).

In the second experiment, we compare with tools supporting string-number conversion on the benchmarks collected from the symbolic executor Py-Conbyte³, which has the supports of string-number conversion. We ran it on several examples collected from the LeetCode platform and from Python core libraries, which involve diverse usages of stringnumber conversion in Python such as parsing date-time, verifying and restoring IP addresses from strings, etc. We also have examples that encode execution paths of some JavaScript programs (the Luhn algorithm and some array manipulations).

All experiments were executed on a machine with 4-core CPU and 16 GiB RAM. The timeout was set to 10s for each test. We use the results from Z3-Trau, CVC4, and Z3 as the reference answer for the validation of the correctness of the results. Occasionally, two of them report inconsistent answers (one SAT and one UNSAT). To decide which solver is right, we developed a validator. It takes the model *I* returned from the solver who reported SAT, assigns I(x) to all variables *x* in the test to obtain a modified test, and re-evaluates the modified test by multiple solvers. If the results from all solvers are consistent, we mark the test SAT or UNSAT according to the results. Otherwise, we manually simplify and inspect the test until we get a conclusive result.

The results of the experiments are summarized in Table 1, Table 2, and Table 3. Rows with heading SAT/UNSAT show numbers of solved formulae. Rows with heading UNKNOWN or TIMEOUT indicate the number of instances for which the solver fails to return an answer. ERROR means system crashes due to various reasons (usually out of memory). INCORRECT shows the number of cases where the tool gives a wrong answer.

Table	1. Result	s of Z3-Tra	.u, CVC4,	Z3, a	nd Z3	3Str3 o	n l	Basic
String	Constrai	nt benchm	arks.					

		Z3-Trau	CVC4	Z3	Z3Str3
	SAT	21377	20350	18492	3037
	UNSAT	3860	3841	3847	3816
DarEar	UNKNOWN	0	0	0	7
FYEX	TIMEOUT	184	1230	3082	16872
	ERROR	0	0	0	1675
	INCORRECT	0	0	0	14
	SAT	877	874	881	661
	UNSAT	1785	1785	1785	1780
LeekCe de	UNKNOWN	0	0	0	122
LeetCode	TIMEOUT	0	7	0	90
	ERROR	4	0	0	13
	INCORRECT	0	0	0	0
	SAT	515	615	338	505
	UNSAT	301	255	198	192
Stain gEurog	UNKNOWN	0	0	0	4
StringFuzz	TIMEOUT	249	195	529	364
	ERROR	0	0	0	0
	INCORRECT	0	0	0	0
	SAT	13	11	12	8
	UNSAT	822	818	808	774
4	UNKNOWN	0	0	0	4
CVC4pred	TIMEOUT	0	6	15	38
	ERROR	0	0	0	11
	INCORRECT	0	0	0	0
	SAT	10	9	7	2
	UNSAT	1032	1026	1022	957
aval	UNKNOWN	0	0	0	3
CVC4 _{term}	TIMEOUT	3	10	16	58
	ERROR	0	0	0	11
	INCORRECT	0	0	0	14
	SAT	22792	21859	19730	4213
	UNSAT	7800	7725	7550	7519
T (1	UNKNOWN	0	0	0	140
Total	TIMEOUT	436	1448	3642	17422
	ERROR	4	0	0	1710
	INCORRECT	0	0	0	28

From Table 1, we can see that the performance of Z3-Trau is as good as that of the most competitive tools such as CVC4 and Z3 on basic string constraints. In all of the benchmarks, Z3-Trau ranked either the 1st or the 2nd on the number of solved (SAT+UNSAT) cases. On the StringFuzz benchmarks that are SAT, Z3-Trau does not perform as well as the best performing tool. We however do not consider this crucial because these benchmarks are just randomly generated for debugging. On the most important benchmarks, those that come from program analysis, Z3-Trau is comparable to the best performing tool.

From Table 2, we can see that Z3-Trau significantly outperforms all the other tools. The second best tool, Z3, fails on 50 times more examples.

As an addition experiment, we have encoded the check-Luhn algorithm introduced in Section 1 for the cases with 2 to 12 loops (digits). We ran these tests with the timeout set

³https://github.com/spencerwuwu/py-conbyte

Table 2. Results of Z3-Trau, CVC4, Z3, and Z3Str3 on String-Number Conversion benchmark.

		Z3-Trau	CVC4	Z3	Z3Str3
	SAT	2501	1659	1993	239
	UNSAT	16394	15604	16124	15288
Tastanda	UNKNOWN	0	0	0	623
Leetcode	TIMEOUT	32	1664	810	2337
	ERROR	0	0	0	332
	INCORRECT	0	0	0	108
	SAT	1922	560	1839	206
	UNSAT	724	666	724	642
DuthonLib	UNKNOWN	0	0	0	45
FytholiLib	TIMEOUT	0	1420	83	1710
	ERROR	0	0	0	41
	INCORRECT	0	0	0	2
	SAT	20	3	16	4
	UNSAT	0	0	0	0
Torra Comint	UNKNOWN	0	9	0	0
Javascript	TIMEOUT	0	8	4	10
	ERROR	0	0	0	6
	INCORRECT	0	0	0	0
	SAT	4443	2222	3848	449
	UNSAT	17118	16270	16848	15930
Tatal	UNKNOWN	0	9	0	668
Total	TIMEOUT	32	3092	897	4057
	ERROR	0	0	0	379
	INCORRECT	0	0	0	110

to 120s. The result is summarized in Table 3. In these tests, Z3-Trau can solve all problems within 1s while CVC4 only returns a model for cases of 2 to 5 loops and Z3Str3 could not solve any of these problems (either TIMEOUT, ERROR, or UNKNOWN). However, Z3 can still solve 5 out of the 11 problems. The behavior of Z3 is not entirely unexpected. All the problems are satisfiable and the solver may be lucky to guess the solution quickly.

Table 3. Comparison of Z3-Trau, CVC4, Z3, and Z3Str3 with checkLuhn problems of 2 to 12 loops.

# of Loops	Z3-Trau	CVC4	Z3	Z3Str3
2	SAT(0.27s)	SAT(0.17s)	SAT(0.11s)	ERROR
3	SAT(0.29s)	SAT(6.94s)	SAT(0.13s)	ERROR
4	SAT(0.37s)	SAT(4.92s)	SAT(0.24s)	ERROR
5	SAT(0.39s)	SAT(30.86s)	SAT(0.13s)	ERROR
6	SAT(0.41s)	TIMEOUT	TIMEOUT	UNKNOWN
7	SAT(0.51s)	TIMEOUT	TIMEOUT	ERROR
8	SAT(0.53s)	TIMEOUT	TIMEOUT	ERROR
9	SAT(0.63s)	TIMEOUT	SAT(0.31s)	ERROR
10	SAT(0.69s)	TIMEOUT	TIMEOUT	TIMEOUT
11	SAT(0.71s)	TIMEOUT	TIMEOUT	ERROR
12	SAT(0.74s)	TIMEOUT	TIMEOUT	ERROR

10 Related Works

To the best of our knowledge, the study of solving string constraint traces back to 1946, when Quine [32] showed that the first-order theory of string equality constraints (a.k.a. word equations) is undecidable. Makanin achieves a milestone [28] by showing that the class of quantifier-free string equality constraints is decidable. Since then, several works, e.g., [19, 20, 29–31, 35, 39], consider the decidability and complexity of different subclasses of string equality constraints.

Satisfiability of string constraints is a challenging problem. The satisfiability of equality constraints combined with length constraints of the form |x| = |y| is already opened for more than 20 years [11]. Numerous decidable fragments were proposed [3, 5, 7, 12, 13, 27]. Among them, the chainfree fragment [5] used by our over-approximation module is the largest known decidable fragment, which allows us to produce more precise over-approximation and hence solve many UNSAT instances efficiently.

The strong practical motivation led to the rise of several string constraints solvers that concentrate on solving practical problem instances. Several tools handle string constraints assuming a fixed upper bound on the length of strings and translate them to boolean satisfiability problems [23, 37, 38]. Our method, on the other hand, allows analyzing constraints without a length limit and still with some completeness guarantees, i.e., within the language defined by PFAs.

More recently, *DPLL(T)-based* string solvers [2, 4, 8, 9, 13, 21, 41, 42, 46] lift the restriction of strings of bounded length. They usually support a variety of string constraints, including all basic string constraints, and sometimes also regular/rational relations. The typical procedure they used for solving equality constraints is to split them into simpler subcases, in combination with powerful techniques for Boolean reasoning to curb the resulting exponential search space. In contrast, our approach uses a completely different search strategy. We restrict the solution space to some predefined pattern and step-wisely enrich the pattern in use.

The most relevant work to ours is the FA-based approach [2] that projects the solution space of variables to a generalization of flat automata. The main difference is that our approach works fully symbolically, which is enabled through using integer variables as characters and hence PFAs. The use of integer variables as characters allows our approach to handle string-number constraints efficiently – the values and number of occurrences of those variables can be directly converted to numbers in a linear formula.

In principle, FA-based solvers can be extended to support string-number conversion too, but PFA are much more efficient. Extending FA-based approach would require multiple if-then-else statements (e.g., saying that if a transition labeled '0' is taken, then the corresponding number is 0), which introduces a significant amount of additional predicates that the DPLL engine needs to evaluate. We have confirmed this in our preliminary experiment.

PASS [26] uses quantified formulae over arrays with symbolic length to encode string constraints, and a specialized quantifier elimination to solve them. Though it differs from our approach significantly and handling the quantification is expensive, PASS is indeed similar in how finite automata and string-number conversion constraints are translated to formulae. Our translation is in general more efficient because it uses only linear arithmetics. In contrast, PASS translates string constraints to formulae with quantifier and array predicates. We did not compare with PASS empirically. The authors informed us that PASS is no longer maintained and is not open-source.

A further direction is *automata-based* solvers for analyzing string-manipulated programs. ABC [6] and Stranger [44] soundly over-approximates string constraints using multitape automata [45], and outperforms DPLL(T)-based solvers when checking single execution traces, according to some evaluations [22]. People also studied the combination of automata-based algorithms with with model checking algorithms, in particular, IC3/PDR, for more efficient checking of the emptiness for automata [21, 43]. However, many kinds of constraints, including length constraints and word equations, cannot be entirely handled by automata-based solvers.

11 Conclusion and Future Works

In this paper, we report a novel approach for solving string constraints with string-number conversion and implemented it as an open-source tool Z3-Trau. For now, it support basic string constraints, string-number conversion, and also operations that can be encoded to them (e.g., CONTAINS, PREFIXOF). Since Z3-Trau is built inside the SMT solver Z3, we also get the power of processing formulae in the combination of different theories (e.g., array). Hence our tool can support the encoding of a wide range of program expressions. There are several avenues for future works. First, we are planning to integrate it with the JavaScript symbolic executor COSETTE [36]. We believe such integration is feasible. We are also planning to merge Z3-Trau with the main branch of the Z3 solver. For technical development, we think it would be interesting to consider the (symbolic) flattening of an even larger set of string operations, such as the one containing REPLACEALL and SPLIT.

Acknowledgments

This research was partially supported by the Swedish Foundation for Strategic Research, Ministry of Science and Technology Project, Taiwan (no. 106-2221-E-001 -009 -MY3), Ministry of Education, Youth and Sports of Czech Republic (project IT4Innovations excellence in science–LQ1602 and project LL1908), and FIT BUT internal grant FIT-S-20-6427.

References

- [1] Parosh Aziz Abdulla, Mohamed Faouzi Atig, Yu-Fang Chen, Bui Phi Diep, Lukás Holík, Ahmed Rezine, and Philipp Rümmer. 2017. Flatten and conquer: a framework for efficient analysis of string constraints. In Proceedings of the 38th ACM SIGPLAN Conference on Programming Language Design and Implementation, PLDI 2017, Barcelona, Spain, June 18-23, 2017, Albert Cohen and Martin T. Vechev (Eds.). ACM, 602–617.
- [2] Parosh Aziz Abdulla, Mohamed Faouzi Atig, Yu-Fang Chen, Bui Phi Diep, Lukás Holík, Ahmed Rezine, and Philipp Rümmer. 2018. Trau: SMT solver for string constraints. In 2018 Formal Methods in Computer

Aided Design, FMCAD 2018, Austin, TX, USA, October 30 - November 2, 2018, Nikolaj Bjørner and Arie Gurfinkel (Eds.). IEEE, 1–5.

- [3] Parosh Aziz Abdulla, Mohamed Faouzi Atig, Yu-Fang Chen, Lukás Holík, Ahmed Rezine, Philipp Rümmer, and Jari Stenman. 2014. String Constraints for Verification. In Computer Aided Verification - 26th International Conference, CAV 2014, Held as Part of the Vienna Summer of Logic, VSL 2014, Vienna, Austria, July 18-22, 2014. Proceedings (Lecture Notes in Computer Science), Armin Biere and Roderick Bloem (Eds.), Vol. 8559. Springer, 150–166.
- [4] Parosh Aziz Abdulla, Mohamed Faouzi Atig, Yu-Fang Chen, Lukás Holík, Ahmed Rezine, Philipp Rümmer, and Jari Stenman. 2015. Norn: An SMT Solver for String Constraints. In *Computer Aided Verification* - 27th International Conference, CAV 2015, San Francisco, CA, USA, July 18-24, 2015, Proceedings, Part I (Lecture Notes in Computer Science), Daniel Kroening and Corina S. Pasareanu (Eds.), Vol. 9206. Springer, 462–469.
- [5] Parosh Aziz Abdulla, Mohamed Faouzi Atig, Bui Phi Diep, Lukás Holík, and Petr Janku. 2019. Chain-Free String Constraints. In Automated Technology for Verification and Analysis - 17th International Symposium, ATVA 2019, Taipei, Taiwan, October 28-31, 2019, Proceedings (Lecture Notes in Computer Science), Yu-Fang Chen, Chih-Hong Cheng, and Javier Esparza (Eds.), Vol. 11781. Springer, 277–293.
- [6] Abdulbaki Aydin, William Eiers, Lucas Bang, Tegan Brennan, Miroslav Gavrilov, Tevfik Bultan, and Fang Yu. 2018. Parameterized model counting for string and numeric constraints. In Proceedings of the 2018 ACM Joint Meeting on European Software Engineering Conference and Symposium on the Foundations of Software Engineering, ESEC/SIGSOFT FSE 2018, Lake Buena Vista, FL, USA, November 04-09, 2018, Gary T. Leavens, Alessandro Garcia, and Corina S. Pasareanu (Eds.). ACM, 400-410.
- [7] Pablo Barceló, Diego Figueira, and Leonid Libkin. 2013. Graph Logics with Rational Relations. *Logical Methods in Computer Science* 9, 3 (2013).
- [8] Clark W. Barrett, Christopher L. Conway, Morgan Deters, Liana Hadarean, Dejan Jovanovic, Tim King, Andrew Reynolds, and Cesare Tinelli. 2011. CVC4. In *Computer Aided Verification - 23rd International Conference, CAV 2011, Snowbird, UT, USA, July 14-20, 2011. Proceedings* (*Lecture Notes in Computer Science*), Ganesh Gopalakrishnan and Shaz Qadeer (Eds.), Vol. 6806. Springer, 171–177.
- [9] Murphy Berzish, Vijay Ganesh, and Yunhui Zheng. 2017. Z3str3: A string solver with theory-aware heuristics. In 2017 Formal Methods in Computer Aided Design, FMCAD 2017, Vienna, Austria, October 2-6, 2017, Daryl Stewart and Georg Weissenbacher (Eds.). IEEE, 55–59.
- [10] Dmitry Blotsky, Federico Mora, Murphy Berzish, Yunhui Zheng, Ifaz Kabir, and Vijay Ganesh. 2018. StringFuzz: A Fuzzer for String Solvers. In Computer Aided Verification - 30th International Conference, CAV 2018, Held as Part of the Federated Logic Conference, FloC 2018, Oxford, UK, July 14-17, 2018, Proceedings, Part II (Lecture Notes in Computer Science), Hana Chockler and Georg Weissenbacher (Eds.), Vol. 10982. Springer, 45–51.
- [11] J. Richard Büchi and Steven Senger. 1988. Definability in the Existential Theory of Concatenation and Undecidable Extensions of this Theory. *Math. Log. Q.* 34, 4 (1988), 337–342.
- [12] Taolue Chen, Yan Chen, Matthew Hague, Anthony W. Lin, and Zhilin Wu. 2018. What is decidable about string constraints with the ReplaceAll function. *PACMPL* 2, POPL (2018), 3:1–3:29.
- [13] Taolue Chen, Matthew Hague, Anthony W. Lin, Philipp Rümmer, and Zhilin Wu. 2019. Decision procedures for path feasibility of stringmanipulating programs with complex operations. *PACMPL* 3, POPL (2019), 49:1–49:30.
- [14] Joel D. Day, Vijay Ganesh, Paul He, Florin Manea, and Dirk Nowotka. 2018. The Satisfiability of Word Equations: Decidable and Undecidable Theories. In Reachability Problems - 12th International Conference, RP 2018, Marseille, France, September 24-26, 2018, Proceedings (Lecture Notes

in Computer Science), Igor Potapov and Pierre-Alain Reynier (Eds.), Vol. 11123. Springer, 15–29.

- [15] Leonardo Mendonça de Moura and Nikolaj Bjørner. 2008. Z3: An Efficient SMT Solver. In Tools and Algorithms for the Construction and Analysis of Systems, 14th International Conference, TACAS 2008, Held as Part of the Joint European Conferences on Theory and Practice of Software, ETAPS 2008, Budapest, Hungary, March 29-April 6, 2008. Proceedings (Lecture Notes in Computer Science), C. R. Ramakrishnan and Jakob Rehof (Eds.), Vol. 4963. Springer, 337–340.
- [16] ECMA ECMAScript, European Computer Manufacturers Association, et al. 2019. Ecmascript language specification. https://www.ecmainternational.org/ecma-262/
- [17] Ting Gan, Mingshuai Chen, Liyun Dai, Bican Xia, and Naijun Zhan. 2015. Decidability of the Reachability for a Family of Linear Vector Fields. In Automated Technology for Verification and Analysis - 13th International Symposium, ATVA 2015, Shanghai, China, October 12-15, 2015, Proceedings (Lecture Notes in Computer Science), Bernd Finkbeiner, Geguang Pu, and Lijun Zhang (Eds.), Vol. 9364. Springer, 482–499.
- [18] Ting Gan, Mingshuai Chen, Yangjia Li, Bican Xia, and Naijun Zhan. 2018. Reachability Analysis for Solvable Dynamical Systems. *IEEE Trans. Automat. Contr.* 63, 7 (2018), 2003–2018.
- [19] Vijay Ganesh and Murphy Berzish. 2016. Undecidability of a Theory of Strings, Linear Arithmetic over Length, and String-Number Conversion. *CoRR* abs/1605.09442 (2016). arXiv:1605.09442 http: //arxiv.org/abs/1605.09442
- [20] Vijay Ganesh, Mia Minnes, Armando Solar-Lezama, and Martin C. Rinard. 2012. Word Equations with Length Constraints: What's Decidable?. In Hardware and Software: Verification and Testing - 8th International Haifa Verification Conference, HVC 2012, Haifa, Israel, November 6-8, 2012. Revised Selected Papers (Lecture Notes in Computer Science), Armin Biere, Amir Nahir, and Tanja E. J. Vos (Eds.), Vol. 7857. Springer, 209–226.
- [21] Lukás Holík, Petr Janku, Anthony W. Lin, Philipp Rümmer, and Tomás Vojnar. 2018. String constraints with concatenation and transducers solved efficiently. *PACMPL* 2, POPL (2018), 4:1–4:32.
- [22] Scott Kausler and Elena Sherman. 2014. Evaluation of string constraint solvers in the context of symbolic execution. In ACM/IEEE International Conference on Automated Software Engineering, ASE '14, Vasteras, Sweden - September 15 - 19, 2014, Ivica Crnkovic, Marsha Chechik, and Paul Grünbacher (Eds.). ACM, 259–270.
- [23] Adam Kiezun, Vijay Ganesh, Philip J. Guo, Pieter Hooimeijer, and Michael D. Ernst. 2009. HAMPI: a solver for string constraints. In Proceedings of the Eighteenth International Symposium on Software Testing and Analysis, ISSTA 2009, Chicago, IL, USA, July 19-23, 2009, Gregg Rothermel and Laura K. Dillon (Eds.). ACM, 105–116.
- [24] Zachary Kincaid, Jason Breck, John Cyphert, and Thomas W. Reps. 2019. Closed forms for numerical loops. *PACMPL* 3, POPL (2019), 55:1–55:29.
- [25] Leetcode. 2019. LeetCode. https://leetcode.com/
- [26] Guodong Li and Indradeep Ghosh. 2013. PASS: String Solving with Parameterized Array and Interval Automaton. In Hardware and Software: Verification and Testing - 9th International Haifa Verification Conference, HVC 2013, Haifa, Israel, November 5-7, 2013, Proceedings (Lecture Notes in Computer Science), Valeria Bertacco and Axel Legay (Eds.), Vol. 8244. Springer, 15–31.
- [27] Anthony Widjaja Lin and Pablo Barceló. 2016. String solving with word equations and transducers: towards a logic for analysing mutation XSS. In Proceedings of the 43rd Annual ACM SIGPLAN-SIGACT Symposium on Principles of Programming Languages, POPL 2016, St. Petersburg, FL, USA, January 20 - 22, 2016, Rastislav Bodík and Rupak Majumdar (Eds.). ACM, 123–136.
- [28] Gennadiy Semenovich Makanin. 1977. The problem of solvability of equations in a free semigroup. *Matematicheskii Sbornik* 145, 2 (1977),

147-236.

- [29] Yuri Matiyasevich. 2008. Computation Paradigms in Light of Hilbert's Tenth Problem. In New Computational Paradigms: Changing Conceptions of What is Computable, S. Barry Cooper, Benedikt Löwe, and Andrea Sorbi (Eds.). Springer New York, New York, NY, 59–85.
- [30] Wojciech Plandowski. 1999. Satisfiability of Word Equations with Constants is in PSPACE. In 40th Annual Symposium on Foundations of Computer Science, FOCS '99, 17-18 October, 1999, New York, NY, USA. IEEE Computer Society, 495–500.
- [31] Wojciech Plandowski. 2006. An efficient algorithm for solving word equations. In Proceedings of the 38th Annual ACM Symposium on Theory of Computing, Seattle, WA, USA, May 21-23, 2006, Jon M. Kleinberg (Ed.). ACM, 467–476.
- [32] Willard Van Orman Quine. 1946. Concatenation as a Basis for Arithmetic. J. Symb. Log. 11, 4 (1946), 105–114.
- [33] Andrew Reynolds, Andres Nötzli, Clark W. Barrett, and Cesare Tinelli. 2019. High-Level Abstractions for Simplifying Extended String Constraints in SMT. In Computer Aided Verification - 31st International Conference, CAV 2019, New York City, NY, USA, July 15-18, 2019, Proceedings, Part II (Lecture Notes in Computer Science), Isil Dillig and Serdar Tasiran (Eds.), Vol. 11562. Springer, 23–42.
- [34] Andrew Reynolds, Maverick Woo, Clark W. Barrett, David Brumley, Tianyi Liang, and Cesare Tinelli. 2017. Scaling Up DPLL(T) String Solvers Using Context-Dependent Simplification. In Computer Aided Verification - 29th International Conference, CAV 2017, Heidelberg, Germany, July 24-28, 2017, Proceedings, Part II (Lecture Notes in Computer Science), Rupak Majumdar and Viktor Kuncak (Eds.), Vol. 10427. Springer, 453–474.
- [35] John Michael Robson and Volker Diekert. 1999. On Quadratic Word Equations. In STACS 99, 16th Annual Symposium on Theoretical Aspects of Computer Science, Trier, Germany, March 4-6, 1999, Proceedings (Lecture Notes in Computer Science), Christoph Meinel and Sophie Tison (Eds.), Vol. 1563. Springer, 217–226.
- [36] José Fragoso Santos, Petar Maksimovic, Théotime Grohens, Julian Dolby, and Philippa Gardner. 2018. Symbolic Execution for JavaScript. In Proceedings of the 20th International Symposium on Principles and Practice of Declarative Programming, PPDP 2018, Frankfurt am Main, Germany, September 03-05, 2018, David Sabel and Peter Thiemann (Eds.). ACM, 11:1–11:14.
- [37] Prateek Saxena, Devdatta Akhawe, Steve Hanna, Feng Mao, Stephen McCamant, and Dawn Song. 2010. A Symbolic Execution Framework for JavaScript. In 31st IEEE Symposium on Security and Privacy, S&P 2010, 16-19 May 2010, Berleley/Oakland, California, USA. IEEE Computer Society, 513–528.
- [38] Prateek Saxena, Steve Hanna, Pongsin Poosankam, and Dawn Song. 2010. FLAX: Systematic Discovery of Client-side Validation Vulnerabilities in Rich Web Applications. In Proceedings of the Network and Distributed System Security Symposium, NDSS 2010, San Diego, California, USA, 28th February - 3rd March 2010. The Internet Society.
- [39] Klaus U. Schulz. 1990. Makanin's Algorithm for Word Equations - Two Improvements and a Generalization. In Word Equations and Related Topics, First International Workshop, IWWERT '90, Tübingen, Germany, October 1-3, 1990, Proceedings (Lecture Notes in Computer Science), Klaus U. Schulz (Ed.), Vol. 572. Springer, 85–150.
- [40] Helmut Seidl, Thomas Schwentick, and Anca Muscholl. 2008. Counting in trees. In Logic and Automata: History and Perspectives [in Honor of Wolfgang Thomas] (Texts in Logic and Games), Jörg Flum, Erich Grädel, and Thomas Wilke (Eds.), Vol. 2. Amsterdam University Press, 575– 612.
- [41] Minh-Thai Trinh, Duc-Hiep Chu, and Joxan Jaffar. [n.d.]. S3: A Symbolic String Solver for Vulnerability Detection in Web Applications. In Proceedings of the 2014 ACM SIGSAC Conference on Computer and Communications Security, Scottsdale, AZ, USA, November 3-7, 2014, Gail-Joon Ahn, Moti Yung, and Ninghui Li (Eds.). ACM, 1232–1243.

Efficient Handling of String-Number Conversion

- [42] Minh-Thai Trinh, Duc-Hiep Chu, and Joxan Jaffar. 2016. Progressive Reasoning over Recursively-Defined Strings. In Computer Aided Verification - 28th International Conference, CAV 2016, Toronto, ON, Canada, July 17-23, 2016, Proceedings, Part I (Lecture Notes in Computer Science), Swarat Chaudhuri and Azadeh Farzan (Eds.), Vol. 9779. Springer, 218–240.
- [43] Hung-En Wang, Tzung-Lin Tsai, Chun-Han Lin, Fang Yu, and Jie-Hong R. Jiang. 2016. String Analysis via Automata Manipulation with Logic Circuit Representation. In Computer Aided Verification - 28th International Conference, CAV 2016, Toronto, ON, Canada, July 17-23, 2016, Proceedings, Part I (Lecture Notes in Computer Science), Swarat Chaudhuri and Azadeh Farzan (Eds.), Vol. 9779. Springer, 241–260.
- [44] Fang Yu, Muath Alkhalaf, and Tevfik Bultan. 2010. Stranger: An Automata-Based String Analysis Tool for PHP. In Tools and Algorithms for the Construction and Analysis of Systems, 16th International Conference, TACAS 2010, Held as Part of the Joint European Conferences on Theory and Practice of Software, ETAPS 2010, Paphos, Cyprus, March 20-28, 2010. Proceedings (Lecture Notes in Computer Science), Javier

Esparza and Rupak Majumdar (Eds.), Vol. 6015. Springer, 154-157.

- [45] Fang Yu, Ching-Yuan Shueh, Chun-Han Lin, Yu-Fang Chen, Bow-Yaw Wang, and Tevfik Bultan. 2016. Optimal sanitization synthesis for web application vulnerability repair. In Proceedings of the 25th International Symposium on Software Testing and Analysis, ISSTA 2016, Saarbrücken, Germany, July 18-20, 2016, Andreas Zeller and Abhik Roychoudhury (Eds.). ACM, 189–200.
- [46] Yunhui Zheng, Vijay Ganesh, Sanu Subramanian, Omer Tripp, Murphy Berzish, Julian Dolby, and Xiangyu Zhang. 2017. Z3str2: an efficient solver for strings, regular expressions, and length constraints. *Formal Methods Syst. Des.* 50, 2-3 (2017), 249–288.
- [47] Yunhui Zheng, Xiangyu Zhang, and Vijay Ganesh. 2013. Z3-str: a z3based string solver for web application analysis. In *Joint Meeting of the European Software Engineering Conference and the ACM SIGSOFT Symposium on the Foundations of Software Engineering, ESEC/FSE'13, Saint Petersburg, Russian Federation, August 18-26, 2013, Bertrand Meyer,* Luciano Baresi, and Mira Mezini (Eds.). ACM, 114–124.